Author’s response to reviews

Title: Short- and long-term outcomes in infective endocarditis patients: a systematic review and meta-analysis

Authors:
Tadesse Melaku Abegaz (abegaztadessa981@gmail.com)
Akshaya Srikanth Bahagavathula (akshaypharmd@gmail.com)
Eyob Alemayehu Gebreyohans (justeyob@gmail.com)
Alemayehu Berhane Mekonenn (aber5592@uni.sydney.edu.au)
Tamrat Befekadu Abebe (befekadutamrat@gmail.com)

Version: 2 Date: 29 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for considering our paper for publication.

we respond for the issues related to the questions and comments given for us as follows.

1. the role of the author named TMA is substantial. accordingly, we included the contribution of this regarded author in the authors contribution section.

2. we included the email address of all authors in the first page.

3. 'additional files' section is added next to references and figures

4. the final version of the manuscript is uploaded as a clean document.